Author's response to reviews

Title: Spatial analysis of hemorrhagic fever with renal syndrome in China

Authors:

Liqun Fang (fang_lq@163.com)
Lei Yan (yanlei@188.com)
Song Liang (liang1@berkeley.edu)
Sake Jan de Vlas (s.devlas@erasusmc.nl)
Dan Feng (fddd@263.net)
Xiaona Han (cynthia817@126.com)
Wenjuan Zhao (vivien2082@yahoo.com.cn)
Bing Xu (bing.xu@geog.utah.edu)
Ling Bian (bian@geog.buffalo.edu)
Hong Yang (anni_yang@163.com)
Peng Gong (gong@irsac.ac.cn)
Jan Hendrik Richardus (j.richardus@erasusmc.nl)
Wuchun Cao (caowc@ic.bmi.ac.cn)

Version: 5 Date: 25 April 2006

Author's response to reviews:

Dear Editor,

Thank you for the work about our manuscript.

But because of my carelessness, there are two places with incorrect formatting. I have revised them and list them as follows. Please correct them before publication.

(1) In the line of authors of the title page, there is another unnecessary blank before the author "Dan Feng".
(2) In the page of table, the table 1 was divided into two pages. I have adjusted it to one page.

In addition, I have uploaded a revised manuscript and please check it.

Many thanks in advance!

With best wishes,

Liqun Fang